\documentclass[10pt,letterpaper, final]{report} % \documentclass{} required in all files

\usepackage{../preamble/myWEPS} % This is now our single preamble file inclusion (it assumes a “.sty” file extension)

\begin{document}

WEPS provides numerous outputs to aid the user in conservation planning. These outputs are accessed through
the \textbf{\ul{V}iewOutput} menu on the main WEPS screen (\cref{fig:WEPS_main_screen})
with some reports also being accessible via the button bar.
Clicking on this menu displays two choices, \textbf{Most Recent Run} and \textbf{Previous Run}.
Clicking on \textbf{Most Recent Run} displays a list of output options for the most recent (usually last) WEPS Run.
The \textbf{Previous Run} choice allows the user to view results of previous WEPS Runs.
A description of the choices under these two sub-menus are shown in \cref{fig:Report_Menu_Options}.

\begin{figure}[H]
	\qquad
		\subcaptionbox{Most Recent Run Menu Options}{\includegraphics[width=0.40\linewidth,scale=3.0]{images/Reports/Most_Recent_Run_menu.jpg}}%
	\qquad\qquad\qquad
		\subcaptionbox{Previous Run Menu Options}{\includegraphics[width=0.40\linewidth,scale=3.0]{images/Reports/Previous_Run_menu.jpg}}%
		\caption{Report Menu Options.}
	\label{fig:Report_Menu_Options}
\end{figure}

For each of those same reports there is a button bar included at the top of the display screen
(\cref{fig:WEPS_Output_Reports_button bar}).

\begin{figure}[H]
\centering
	\includegraphics[scale=2.75]{images/Reports/output_reports_button_bar.jpg}
\caption{WEPS Output Reports button bar.}
\label{fig:WEPS_Output_Reports_button bar}
\end{figure}

\begin{itemize}

\item[]{\scalerel*{\includegraphics{images/icons/print.png}}{\strut} This button allows the user to print the report.}

\item[]{\scalerel*{\includegraphics{images/icons/Exportbutton.png}}{\strut}
This button allows the user to create a file of the report in any of the following formats:
a) PDF (Adobe Postscript), b) RTF (Rich Text Format), c) ODT (Open Document Format),
d) HTML (Hypertext Markup Language), e) CSV (Comma Separated Values),
f) XML (Extensible Markup Language), JasperReports and h) Embedded Images XML.}

\item[]{\scalerel*{\includegraphics{images/icons/adobe.png}}{\strut}
This button allows the user to view the report using the operating system's default PDF viewer
to display the PDF generated version of the report.
If the selected report does not have a PDF version available,
a pop-up message will alert the user to that fact.
However, the user can use the previous button bar button
{\scalerel*{\includegraphics{images/icons/Exportbutton.png}}{\strut}
to generate the PDF file manually, if desired}}.

\item[]{\scalerel*{\includegraphics{images/icons/1st_page.png}}{\strut}
This button allow the user to jump back to the 1st page of the report.}

\item[]{\scalerel*{\includegraphics{images/icons/previous_page.png}}{\strut}
This button allows the user to go to the previous page of the report.}

\item[]{\scalerel*{\includegraphics{images/icons/next_page.jpg}}{\strut}
This button allows the user to go to the next page of the report.}

\item[]{\scalerel*{\includegraphics{images/icons/last_page.png}}{\strut}
This button allow the user to go to the last page of the report.}

\item[]{\scalerel*{\includegraphics{images/icons/page_no_field.png}}{\strut}
This button allows the user to go to the specified page number entered into the field.}

\item[]{\scalerel*{\includegraphics{images/icons/new.png}}{\strut}
This button allows the user to resize the report display page to the default (100$\%$) size.}

\item[]{\scalerel*{\includegraphics{images/icons/shrink_page.png}}{\strut}
This button allows the user to resize the report to fit the page vertically in the window.}

\item[]{\scalerel*{\includegraphics{images/icons/enlarge_page.jpg}}{\strut}
This button allows the user to resize the report to fit the page horizontally in the window.}

\item[]{\scalerel*{\includegraphics{images/icons/magnify_25.jpg}}{\strut}
This button allows the user to magnify the report display page by 25$\%$ increments.}

\item[]{\scalerel*{\includegraphics{images/icons/shrink_25.png}}{\strut}
This button allows the user to shrink the report display page by 25$\%$ increments.}

\item[]{\scalerel*{\includegraphics{images/icons/display_size.jpg}}{\strut}
This button displays the current size of the report display page
and allows the user to specify the desired size of the report display page
or to select a specified size through the drop-down arrow button
which provides a choice list menu of pre-selected sizes ranging from 50$\%$ to 800$\%$.}

\end{itemize}

The remaining buttons provide quick access to a subset of the WEPS Run reports for the selected WEPS Run:

\begin{itemize}
\item[]{\scalerel*{\includegraphics{images/icons/ReportR.png}}{\strut}
This button allows the user to access the current WEPS Run Summary Report.}
\item[]{\scalerel*{\includegraphics{images/icons/ReportM.png}}{\strut}
This button allows the user to access the current WEPS Management Rotation Summary Report.}
\item[]{\scalerel*{\includegraphics{images/icons/ReportC.png}}{\strut}
This button allows the user to access the current WEPS Run Crop Summary Report.}
\item[]{\scalerel*{\includegraphics{images/icons/covercrop.png}}{\strut}
This button allows the user to access the current WEPS Run Cover Crop Summary Report.}
\item[]{\scalerel*{\includegraphics{images/icons/cropinterval.png}}{\strut}
This button allows the user to access the current WEPS Run Crop Interval Summary Report.}
\item[]{\scalerel*{\includegraphics{images/icons/ReportS.png}}{\strut}
This button allows the user to access the current WEPS Run Soil Report.}
\item[]{\scalerel*{\includegraphics{images/icons/ReportD.png}}{\strut}
This button allows the user to access the current WEPS Run Tabular Detail Report.}
\item[]{\scalerel*{\includegraphics{images/icons/ReportI.png}}{\strut}
This button allows the user to access the current WEPS Run Confidence Interval Report.}
\item[]{\scalerel*{\includegraphics{images/icons/QuickPlot_button_dropdown.png}}{\strut}
This button and dropdown arrow allows the user to access the current set of WEPS Run QuickPlot Reports.
Clicking on the the QuickPlot icon displays the \textbf{Wind Erosion Soil Loss (Creep+Salt, Susp)} report.
The dropdown arrow provides a list of seven QuickPlot reports to choose from.
They are titled:
\textbf{Wind Erosion Soil Loss (Creep+Salt, Susp)},
\textbf{Canopy and Residue Cover},
\textbf{Precipitation and Irrigation},
\textbf{Wind Energy},
\textbf{Total Wind Erosion Soil Loss (Gross, Net)},
and \textbf{Total Above Ground Crop and Residue Biomass}.}
\end{itemize}

All of the regular WEPS reports
(excluding the \textbf{Tabular Detail Report}, the \textbf{QuickPlot Reports}
and the \textbf{Science Model Reports})
contain a \hypertarget{Report_Header}{header} that also includes the following elements (\cref{fig:Report_Header_Elements}):

\begin{figure}[H]
\centering
\begin{multicols}{2}
  \begin{itemize}[noitemsep]
    \item[•]{Name of WEPS Run}
    \item[•]{WEPS Run folder location}
    \item[•]{Client Name}
    \item[•]{Farm No, Tract No and Field No}
    \item[•]{WEPS Run Date}
    \item[•]{Management scenario file name}
    \item[•]{Soil file name}
    \item[]{ } %Empty item to get row spacing in 2nd column to match first column when we have odd number of items.
  \end{itemize}
\end{multicols}
\caption{List of WEPS Output Reports header elements.}
\label{fig:Report_Header_Elements}
\end{figure}

An example of an actual WEPS Report header section is shown in \cref{fig:Report_Header_Section}.

\begin{figure}[H]
\centering
	\includegraphics[scale=2.0]{images/Reports/Report_Header_Section.jpg}
\caption{WEPS Output Reports header section.}
\label{fig:Report_Header_Section}
\end{figure}

\end{document}